Author’s response to reviews

Title: Lab-based and Diagnosis-based Chronic Kidney Disease Recognition and Staging Concordance

Authors:

Clarissa Diamantidis (clarissa.diamantidis@duke.edu)
Sarah Hale (sarah.hale@duke.edu)
Virginia Wang (virginia.wang@duke.edu)
Valerie Smith (valerie.smith@duke.edu)
Sarah Scholle (scholle@ncqa.org)
Matthew Maciejewski (matthew.maciejewski@duke.edu)

Version: 2 Date: 04 Sep 2019

Author’s response to reviews:

September 4, 2019

Jing Zhang
Assistant Editor
BMC Nephrology
RE: Submission of Manuscript BNEP-D-19-00521R1

Dr. Zhang,

Enclosed please find a further revised manuscript, entitled “Lab-based and Diagnosis-based Chronic Kidney Disease Recognition and Staging Concordance”, that adds the Declarations after the Conclusions and before References, outlines Virginia Wang as an editorial board member of BMC Nephrology and adds a statement that all authors have read and approve of this final version.
Thank you for this qualified acceptance and I hope this clean version without track changes is acceptable to BMC Nephrology.

Sincerely,

Matthew L. Maciejewski, Ph.D.
Professor
Research Career Scientist
Department of Population Health Sciences
Center for Health Services Research in Primary Care
Duke University Medical Center
Durham VA Medical Center